Author's response to reviews

Title: Magnesium Deficiency: Effect on Bone Mineral Density in the Mouse Appendicular Skeleton

Authors:

Dr Helen E. Gruber (hgruber@carolinas.org)
Robert K. Rude (RRude60075@aol.com)
Livvy Y. Wei (livvywei@yahoo.com)
Angie Frausto (frausto@usc.edu)
Barbara G. Mills (bmills@usc.edu)
H. JAMES Norton (inorton@carolinas.org)

Version: 2 Date: 5 Feb 2003

Merely updating one reference which is now published.